Author's response to reviews

Title: A randomized controlled trial of acupuncture to treat functional constipation: Design and protocol

Authors:

Xiaohu Xu (hermanxu@163.com)
Guangying Huang (gyhuang@tjh.tjmu.edu.cn)
Cuihong Zheng (chzheng2008@qq.com)
Mingmin Zhang (mmzhang@tjh.tjmu.edu.cn)

Version: 2 Date: 21 July 2014

Author's response to reviews:

Title: A randomized controlled trial of acupuncture to treat functional constipation: Design and protocol

Authors:

XiaoHu Xu (hermanxu@163.com),
GuangYing Huang (gyhuang@tjh.tjmu.edu.cn),
CuiHong Zheng (chzheng2008@qq.com)
MingMin Zhang (mmzhang@tjh.tjmu.edu.cn)

Version: 2 Date: 21 July 2014

Author's response to reviews: see over

Reviewer's report

Title: A randomized controlled trial of acupuncture to treat functional constipation: Design and protocol

Version: 2 Date: 21 July 2014

Reviewer: HYANGSOOK LEE

Reviewer's report:

The authors made a great deal of efforts to revise their protocol which now seems more readily understandable and improved.

However, some minor but important errors or wrongly made corrections are to be modified.

[Minor Essential Revisions]

1. Regarding the questionnaires to be used in this study, the authors provided the original studies which are about English versions of those questionnaires. The references of them should be replaced with the Chinese validation studies of those questionnaires. For example, the reference of Zung’s self-rating

According to the reviewer’s comments, we have provided the references for CHINESE versions of the questionnaires to be used as follows:


2. The reference 17 is actually a Korean study which cannot support the authors’ claim, “acupuncture and moxibustion as effective complementary and alternative medicine (CAM) have been widely use to treat FC in China and clinically adopted in the western world.” Also, ‘use’ should be corrected to ‘used’ along with other typos and errors (e.g. trail should be trial).

We have made this correction according to the reviewer’s comments and typos and grammatical errors were corrected in revision paper.

Level of interest: An article whose findings are important to those with closely related research interests

Quality of written English: Needs some language corrections before being published

Statistical review: No, the manuscript does not need to be seen by a statistician.

Declaration of competing interests:
I declare that I have no competing interests.